Author's response to reviews

Title: The Effect of Major Depression on Participation in Preventive Health Care Activities

Authors:

Scott B Patten (patten@ucalgary.ca)
Jeanne VA Williams (jvawilli@ucalgary.ca)
Dina H Lavorato (lavorato_dina@hotmail.com)
Michael Eliasziw (eliasziw@ucalgary.ca)

Version: 3 Date: 12 March 2009

Author's response to reviews: see over
March 12, 2009

Editorial Office
BMC Public Health

Re. MS: 1259978111241519 The Effect of Major Depression on Participation in Preventive Health Care Activities

Please find attached a revision of this paper. This letter contains a description of changes made in response to the reviews. A summary of changes is presented below.

1. A conclusions section has been added.
2. All authors are now listed in all of the references.
3. The references (#8 and #33) have been fixed.
4. “Introduction” has been changed to “Background.”
5. Unnecessary capitals removed from title.
6. Degrees removed from after authors’ names.
7. The tables have been moved to after the references.
8. I re-read the manuscript and corrected 3 minor typographical errors.

Sincerely,

Scott B. Patten, MD, FRCP(C), PhD
Professor
Department of Community Health Sciences